ORIGINAL PAPER



# Increasing Belief in the Effectiveness of HIV Treatment as Prevention: Results of Repeated, National Surveys of Australian Gay and Bisexual Men, 2013–15

Martin Holt<sup>1</sup> · Toby Lea<sup>1</sup> · Heather-Marie Schmidt<sup>2</sup> · Dean Murphy<sup>1,3</sup> · Marsha Rosengarten<sup>1,4</sup> · David Crawford<sup>5</sup> · Jeanne Ellard<sup>6</sup> · Johann Kolstee<sup>7</sup> · John de Wit<sup>1</sup>

Published online: 23 January 2016 © Springer Science+Business Media New York 2016

**Abstract** We surveyed Australian gay and bisexual men, assessing belief in HIV treatment as prevention (TasP) and support for early treatment. We identified the characteristics of participants who believed in TasP and supported early treatment using multivariate logistic regression. In 2013, 1316 men participated; 1251 participated in 2015. Belief in TasP increased from 2.6 % in 2013 to 13.1 % in 2015 (p < 0.001). The increase was most noticeable among HIV-positive men (from 9.7 % to 46.2 %). Support for early treatment increased from 71.8 % to 75.3 % (p = 0.02). Belief in TasP was associated with being HIVpositive, having a tertiary education, having recent condomless anal intercourse with casual male partners, and ever having taken post-exposure prophylaxis. Support for early HIV treatment was associated with being younger, living in New South Wales and being in paid employment. We recommend continued monitoring of the growing gap in belief about TasP between HIV-positive men and HIVnegative/untested men.

Martin Holt m.holt@unsw.edu.au

- <sup>1</sup> Centre for Social Research in Health, UNSW Australia, Level 2, John Goodsell Building, Sydney, NSW 2052, Australia
- <sup>2</sup> New South Wales Ministry of Health, Sydney, Australia
- <sup>3</sup> National Drug Research Institute, Curtin University, Melbourne, Australia
- <sup>4</sup> Goldsmiths, University of London, London, UK
- <sup>5</sup> Positive Life NSW, Sydney, Australia
- <sup>6</sup> Australian Research Centre in Sex, Health and Society, La Trobe University, Melbourne, Australia
- <sup>7</sup> ACON, Sydney, Australia

**Resumen** Encuestamos a hombres australianos gay y bisexuales, evaluando la creencia en el tratamiento como prevención del VIH (TaSP - por sus siglas en ingles) y apoyo al tratamiento temprano. Se identificaron las características de los participantes que creyeron en (TasP) y tratamiento temprano mediante regresión logística multivariante. En el 2013, participaron 1.316 hombres; en el 2015 participaron 1.251. La creencia en (TasP) aumentó de 2.6 % en el 2013 a 13.1 % en el 2015 (p < 0.001). El aumento fue más notable entre los hombres VIH-positivos (de 9.7 % a 46.2 %). Apoyo al tratamiento temprano aumentó de 71.8 % a 75.3 % (p = 0.02). La creencia en TasP se asoció con ser VIH-positivo, con tener una educación terciaria, con tener coito anal recientemente sin condón con parejas casuales y con haber tomado, por lo menos alguna vez, la profilaxis posexposición. El apoyo al tratamiento temprano del VIH se asoció con ser más joven, con vivir en Nueva Gales del Sur y con tener empleo remunerado. Se recomienda el monitoreo continuo de la creciente brecha acerca de la creencia en TasP entre hombres VIH-positivos y VIH-negativo/no comprobado.

Keywords Acceptability  $\cdot$  Attitudes  $\cdot$  Cross-sectional surveys  $\cdot$  HIV treatment as prevention  $\cdot$  Men who have sex with men

## Introduction

HIV treatment as prevention (TasP) is a population health strategy that involves offering HIV treatment as early as possible to people living with HIV to prevent disease progression and to reduce the chance of onward transmission. The strategy was developed and advocated by researchers after clinical trial results showed that early antiretroviral treatment dramatically reduced HIV transmission within heterosexual couples where one partner was HIV-positive [1, 2]. Interim findings suggest that HIV treatment is also highly effective in preventing transmission within gay male couples where one partner is HIVpositive [3]. International organisations such as the World Health Organization and community-based HIV organisations have subsequently endorsed the wider use of HIV treatment to improve the health of people living with HIV and to prevent onward transmission, as long as treatment remains voluntary and informed [4–6].

Despite the increased policy emphasis on promoting TasP, there has been remarkably little analysis of the acceptability of HIV treatment as prevention to affected communities [7, 8]. By acceptability, we mean research into whether communities have heard about the strategies (awareness), what they understand or know about them (knowledge), and whether they are willing to use the strategies or support others doing so. British research conducted in 2011-12 found that over half of treatmentnaïve HIV-positive people (primarily gay men) were in favour of the early initiation of treatment for HIV prevention [9]. National surveys of Australian gay and bisexual men conducted during 2011-13 found majority support for the early initiation of HIV treatment but very low (<3 %) levels of belief in the effectiveness of TasP [10]. HIV-positive men and their regular partners were more likely to believe that HIV treatment prevented transmission than HIV-negative and untested men. More recent research with gay and bisexual men in Vancouver found that over half believed that HIV treatment reduced the risk of transmission, but knowledge of TasP was inconsistent [11]. Australian qualitative research has found that HIV-positive people not on treatment expressed a range of reservations about TasP [12], including the idea that TasP undermines an individual's right to decide when to start treatment, that the needs of the population are being put before those of individual patients, and that TasP may encourage greater levels of risk-taking. In contrast, serodiscordant couples (where one partner is HIV-positive) were generally relieved at the trial evidence in favour of TasP [13].

In Australia, the benefits of early HIV treatment and having an undetectable viral load have been actively promoted to gay and bisexual men since 2012, most notably in the *Ending HIV* social marketing campaign developed by the community organisation ACON (first released in New South Wales, then other jurisdictions; see http://endinghiv. org.au). The tagline of the campaign is 'Test More + Treat Early + Stay Safe = Ending HIV'. These education activities reflect and support state and national HIV strategies that encourage earlier diagnosis, treatment and the use of HIV treatment for prevention [14–16].

Australian prescribing guidelines were changed at the beginning of 2014, recommending HIV treatment for all HIV-positive people, regardless of CD4 cell count [17]. Australian clinicians are increasingly willing to prescribe HIV treatment early i.e. soon after diagnosis and before a patient's CD4 cell count drops below 500 [18]. We have surveyed Australian gay and bisexual men every 2 years since 2011, assessing attitudes to different HIV prevention strategies, including TasP [10, 19]. This analysis assesses whether attitudes have changed over time and identifies the characteristics of gay and bisexual men who support TasP and early treatment.

#### Method

### **Participants and Procedures**

Data were collected as part of the PrEPARE Project, a study of Australian gay and bisexual men's attitudes to biomedical HIV prevention technologies, including TasP [10, 19]. The UNSW Human Research Ethics Committee approved the study design. National online surveys of gay and bisexual men were conducted in April-May 2011, June–July 2013 and April–May 2015 using NETQ survey software (NetQuestionnaires Nederland BV). The survey was advertised on Facebook (using paid advertising) and email lists aimed at gay and bisexual men. Consenting participants from the 2013 survey were invited by email to participate in the 2015 survey. Potential participants were directed to the survey website, http://prepareproject.csrh. org, which explained the purpose of the study and provided access to the online questionnaire. Participants were eligible if they were at least 18 years old, male, lived in Australia and were gay or bisexual. No incentive was offered for participation.

### Measures

Wherever possible, the same questions were used in each survey round. We included questions about demographics, sexual practices with men, relationships, HIV testing and HIV status, adapted from behavioural surveillance questionnaires [20]. Due to small sample sizes in some jurisdictions, the state/territory variable was simplified in multivariate analyses to four categories: New South Wales, Victoria, Queensland and other states/territories. A 9-item scale developed in the 2011 survey measured personal experience in using condoms (Cronbach's  $\alpha = 0.88-0.90$ , 2011–15). The scale score was a mean of the items within the scale (from 1 to 5); men with a score of  $\geq$ 4 were classified as having a positive experience in using condoms [21].

The primary outcomes for this analysis were two previously published scales identified after the 2013 survey [10], 'HIV treatment prevents transmission' (3 items, Cronbach's  $\alpha = 0.73$  in 2015) and 'early HIV treatment is necessary' (3 items,  $\alpha = 0.67$  in 2015). Scale items were Likert-type attitudinal items (each scored from 1 = strongly disagree to 5 = strongly agree). Scale scores were a mean of the items within the scale (from 1 to 5), with a score of  $\geq 4$  indicating positive agreement e.g. participants who scored  $\geq 4$  on the HIV treatment prevents transmission scale were classified as agreeing that HIV treatment prevents transmission. This was the scoring method used in our previous publication about attitudes to TasP [10].

#### Data analysis

Before analysis, the database was inspected manually for duplicate or suspicious entries (those with identical IP addresses, responses or email addresses). Stata Version 13.0 (StataCorp LP, USA) was used for data analysis. Statistical significance was set at p < 0.05. Only data from the 2013 and 2015 survey rounds were included in this analysis, as the primary outcome measures (scales) were not included in the 2011 survey. The characteristics of the 2013 and 2015 samples were compared using  $\chi^2$  tests and *t* tests to identify potentially confounding factors, focusing on demographic criteria (e.g. age, education level) and variables we have previously found to be associated with attitudes to TasP e.g. participant's HIV status, partner's HIV status, sexual practices in the previous 6 months, previous use of HIV post-exposure prophylaxis (PEP) [10].

To assess change over time, multivariate logistic regression was used to compare the proportions of men in 2013 and 2015 who agreed that HIV treatment prevents transmission and that early HIV treatment is necessary, controlling for potentially confounding factors (social and behavioural differences between the 2013 and 2015 samples). Multivariate logistic regression was also used to identify the characteristics of men in 2015 who agreed that HIV treatment prevents transmission and the characteristics of men who agreed that early HIV treatment is necessary. To build the multivariate models, we initially identified bivariate associations with the scale scores using logistic regression, reporting unadjusted odds ratios. For these analyses, we initially included demographic variables, sexual practice and relationship variables, HIV status and sexual health testing and variables we have previously found to be associated with attitudes to HIV treatments, such as the previous use of PEP [10, 19]. All variables with a statistically significant bivariate relationship with the scales were block entered into the multivariate models and adjusted odds ratios were calculated.

## Results

In 2013, the online survey was completed by 1316 men, of whom 966 reported that they were HIV-negative, 257 were untested or of unknown HIV status and 93 were HIV-positive. In 2015, the online survey was completed by 1251 men, of whom 990 were HIV-negative, 155 were untested or of unknown HIV status and 106 were HIV-positive. No duplicate or suspicious entries were found. Sample characteristics are shown in Table 1.

Compared to the 2013 sample, men in the 2015 sample were older, more likely to identify as gay, be born in Australia, live outside of a capital city, to have had anal intercourse with regular or casual partners in the previous 6 months, and to have ever used PEP (see Table 1). Across both survey years, previous PEP use was more likely to be reported by HIV-negative men (18.7 %), compared with HIV-positive (14.1 %) and untested/unknown status men (1.2 %; p < 0.001). Men in the 2015 sample were less likely have completed any tertiary education or be a student and more likely to be unemployed or retired. We controlled for these differences in subsequent analyses. The two primary outcome measures (belief that HIV treatment prevents transmission and belief that early HIV treatment is necessary) were slightly correlated with each other in 2013 (r = -0.08, p = 0.003) and 2015 (r = 0.07, p = 0.02).

#### **HIV Treatment Prevents Transmission**

In 2013, 2.6 % of 1316 participants were classified as agreeing that HIV treatment prevents transmission. This rose to 13.1 % of 1251 men in 2015 [adjusted odds ratio (AOR) = 2.57, 95 % confidence interval (CI) 2.07–3.20, p < 0.001, comparing the 2015 sample to the 2013 sample, controlling for confounders]. The increase was most pronounced among HIV-positive men (9.7–46.2 %; AOR = 4.02, 95 % CI 2.28–7.10, p < 0.001) compared with HIV-negative and untested men (2.0–10.0 %; AOR = 2.36, 95 % CI 1.86–3.00, p < 0.001).

The characteristics of men who agreed in 2015 that HIV treatment prevents transmission versus those who did not are shown in Table 2. Belief in TasP was unrelated to sexual identity, country of birth or employment status (analyses not shown). Multivariate logistic regression indicated that agreement that HIV treatment prevents transmission was significantly associated with being HIV-positive, having completed any tertiary education, reporting any condomless anal intercourse with casual male partners in the previous 6 months, and having ever used PEP. Men who lived in Queensland or a regional or rural area of their state or territory were less likely to agree that HIV treatment prevented transmission.

#### Table 1 Sample characteristics

	2013 sample (n = 1316), %	2015 sample (n = 1251), %	t or $\chi^2$ (p value)
Mean age in years (SD)	32.0 (11.1)	33.7 (11.9)	3.40 (<0.001)
Sexual identity			33.81 (<0.001)
Gay	91.6	96.9	
Bisexual	6.5	2.6	
Other	2.0	0.5	
HIV status			24.77 (<0.001)
HIV-negative	73.4	79.1	
Untested/unknown status	19.5	12.4	
HIV-positive	7.1	8.5	
Country of birth			5.01 (0.03)
Australia	77.1	80.7	
Overseas	22.9	19.3	
State or territory			14.58 (0.04)
Australian Capital Territory	3.6	3.4	
New South Wales	31.8	33.7	
Northern Territory	0.4	1.0	
Queensland	20.1	19.3	
South Australia	5.6	7.3	
Tasmania	1.2	1.8	
Victoria	29.1	24.5	
Western Australia	8.2	9.1	
Residential location			15.85 (<0.001)
Capital city	75.8	68.7	
Other city	10.1	12.7	
Regional/rural area	14.1	18.5	
Employment			50.13 (<0.001)
Full-time	55.9	59.1	. ,
Part-time	10.2	11.4	
Student	24.2	14.3	
Unemployed/retired/other	9.7	15.2	
Education	2.17	10.2	11.41 (0.001)
Up to Year 12 (high school)	28.4	34.6	
Trade certificate or university degree	71.6	65.4	
No. of male sex partners (pa	st 6 months)		3.56 (0.17)
0–1	34.8	35.2	
2-10	46.8	43.8	
>10	18.4	21.0	
HIV status of regular male p	bartner		2.65 (0.44)
No regular partner	44.2	42.8	~ /
HIV-negative	43.8	44.2	
Untested/unknown status	7.4	7.1	
HIV-positive	4.6	5.9	
Anal intercourse with regula			14 73 (0.001)
No partners/no intercourse	35.1	32.1	1.75 (0.001)

 Table 1
 Sample characteristics

	2013 sample (n = 1316), %	2015 sample (n = 1251), %	t or $\chi^2$ (p value)
Consistent condom use	17.8	13.7	
Any anal intercourse without condoms	47.1	54.2	
Anal intercourse with casua	past 6 months)	29.08 (<0.001)	
No partners/no intercourse	42.4	40.0	
Consistent condom use	31.1	24.3	
Any anal intercourse without condoms	26.5	35.7	
Ever received post-exposur	EP)	6.04 (0.01)	
No	86.2	82.7	
Yes	13.8	17.3	
Personal experience in usin	1.94 (0.16)		
Negative/neutral (score < 4)	92.6	94.0	
Positive (score $\geq 4$ )	7.4	6.0	

t t test,  $\chi^2$  Chi square test, p probability, SD standard deviation

\* 9-item scale scored from 1 to 5

#### Early HIV Treatment is Necessary

In 2013, 71.8 % of participants were classified as agreeing that early HIV treatment is necessary and this increased to 75.3 % in 2015 (AOR = 1.13, 95 % CI 1.02-1.24, p = 0.02, comparing the 2015 sample to the 2013 sample, controlling for confounders). The increase was concentrated among HIV-positive men (40.9-60.4 %; AOR = 1.42, 95 % CI 1.03–1.97, p = 0.03) rather than HIV-negative and untested men (74.2-76.7 %: AOR = 1.10, 95 % CI 0.99–1.22, p = 0.07).

The characteristics of men who agreed in 2015 that early HIV treatment is necessary versus those who did not are shown in Table 3. Support for early treatment was unrelated to sexual identity, country of birth, residential location, number of male sex partners, anal intercourse with regular or casual partners, PEP use or the participant's experience in using condoms (analyses not shown). Younger men were more likely to agree with the need for early HIV treatment. Belief that early HIV treatment is necessary was less likely among men who lived in Queensland or Victoria (compared with New South Wales) and those who were unemployed or retired.

## Discussion

Our national surveys of Australian gay and bisexual men show that attitudes to HIV treatment for prevention and the early initiation of treatment have become more positive over the last 2 years. This coincides with the introduction of community education campaigns to promote the health and preventative benefits of HIV treatment, a greater focus on TasP in local HIV strategies and changes in prescribing guidelines to support early treatment.

The majority of Australian gay and bisexual men remain sceptical about TasP, but the proportion of men who agree that HIV treatment prevents transmission has increased from 3 % in 2013 to 13 % in 2015. The increase is most pronounced among HIV-positive men, among whom nearly half agree that treatment prevents transmission. A similar level of support has been observed in Canada and the UK [9, 11]. This suggests that Australian HIV-positive men may be more aware of the results from key trials, by exposure to community campaigns or through clinicians, for example. This is supported by the finding that belief in TasP was greater among men who lived in New South Wales and Victoria, the two states with the largest populations of gay and bisexual men and (arguably) more intensive promotion of the benefits of HIV treatment over the last 2 years. In contrast, men in regional and rural areas were less likely to believe in TasP, presumably because they are less likely to access or be exposed to discussion or information about TasP.

Consistent with our previous research [10], we found that men at higher risk of HIV, evidenced by condomless sex with casual partners, the use of PEP, having an HIVpositive partner or a greater number of partners overall, were more likely to agree that HIV treatment prevents transmission. There is a range of possible explanations for this finding. Since the advent of combination antiretroviral treatment, men who are at higher risk of HIV have been consistently found to be more optimistic about the capacity to avoid HIV [7, 22]. This optimism may be accurate (effective treatments do reduce HIV transmission) but it may also be misplaced (HIV-negative and untested men rarely limit sex to diagnosed HIV-positive partners on treatment, so HIV treatment cannot be relied upon to consistently protect them). The experience of PEP, which is generally effective in preventing HIV if the course of drugs is adhered to [23], may foster greater confidence in the effectiveness of HIV treatments in preventing transmission. It is possible that as pre-exposure prophylaxis (PrEP) becomes more available that it will have a similar effect in boosting confidence about TasP. Gay and bisexual men with HIV-positive partners may be more likely to believe in TasP because they have greater exposure to knowledge of HIV treatments and prevention science. Qualitative research also suggests that men in serodiscordant couples are reassured by scientific findings about the efficacy of TasP, giving them greater confidence in managing the risk of transmission within their relationships [13].

In 2013 we found that a majority of Australian gay and bisexual men (72 %) supported the idea of early HIV treatment and this modestly increased to 75 % in 2015. The increase was largely concentrated among HIV-positive men, who in 2013 were much less supportive of early treatment than HIV-negative and untested men. In fact, in contrast to our previous research [10], we no longer find that HIV-positive men or their partners are less likely to support early treatment. This suggests that concerns about early treatment (such as perceiving little personal benefit of early treatment or feeling compelled to start treatment to protect others) have been allayed to some degree among HIV-positive men and their partners, echoing the evolving views of HIV clinicians [12, 13, 18]. Consistent with our previous research [10], younger men were more supportive of early treatment. We also observed that men in New South Wales, where the Ending HIV campaign originated, were much more likely to support the idea of early treatment than men residing in Queensland or Victoria, perhaps reflecting the period of time the campaign has been active in the respective states. Men who were unemployed or retired were less likely than men in full-time employment to support early treatment, perhaps reflecting concerns about the cost of medicines for those on lower incomes.

Our results are limited in a number of ways. We used cross-sectional surveys that could not identify causal processes or analyse changes in the attitudes of the same men over time. As in previous years, we did not give participants any background information about the concept of TasP or current recommendations about when to initiate treatment. Neither did we assess participants' exposure to education campaigns, debates or sources of information about HIV treatment (so our observations about the impact of campaigns are speculative). The profile of participants we recruited is similar to community samples of Australian gay and bisexual men in which HIV risk tends to be elevated [20], but is not a representative sample of gay and bisexual men in Australia [24]. The level of PEP use we observed in our sample appears high, but is similar to levels of lifetime use observed in other studies [25], perhaps reflecting that PEP has been widely available in Australia since 2001 [23].

Our results have a number of implications. They show that Australian gay and bisexual men remain largely supportive of the early initiation of HIV treatment, but most remain sceptical about the use of TasP. As we have noted before, this suggests greater confidence in the personal

1569

	Disagreed/neutral $(n = 1087), n (\%)$	Agreed (n = 164), n (%)	Unadjusted OR (95 % CI)	р	Adjusted OR (95 % CI)	р
Mean age in years (SD)	33.2 (11.9)	36.9 (11.6)	1.02 (1.01-1.03)	< 0.001	1.00 (0.98–1.01)	0.70
HIV status						
HIV-negative	81.0	67.1	Ref.		Ref.	
Untested/unknown status	13.8	3.0	0.27 (0.11-0.66)	0.005	0.52 (0.20-1.35)	0.18
HIV-positive	5.2	29.9	6.88 (4.47-10.57)	< 0.001	6.22 (3.74–10.34)	< 0.001
State or territory						
New South Wales	32.7	40.2	Ref.		Ref.	
Queensland	20.3	12.2	0.49 (0.29-0.83)	0.007	0.49 (0.27-0.88)	0.02
Victoria	23.4	32.3	1.12 (0.76–1.67)	0.57	0.93 (0.60-1.44)	0.75
Other states/territories	23.6	15.2	0.52 (0.32-0.85)	0.009	0.62 (0.37-1.06)	0.08
Residential location						
Capital city	66.1	86.0	Ref.		Ref.	
Other city	13.9	4.9	0.27 (0.13-0.56)	< 0.001	0.52 (0.24–1.12)	0.09
Regional/rural area	20.0	9.1	0.35 (0.20-0.61)	< 0.001	0.44 (0.24-0.80)	0.007
Education						
Up to Year 12 (high school)	36.6	21.3	Ref.		Ref.	
Trade certificate or university degree	63.4	78.7	2.13 (1.44–3.16)	< 0.001	1.49 (0.96–2.31)	0.08
No. of male sex partners in pa	st 6 months					
0–1	38.3	14.6	Ref.		Ref.	
2-10	43.1	48.8	2.96 (1.84-4.76)	< 0.001	1.37 (0.64-2.95)	0.41
>10	18.7	36.6	5.12 (3.10-8.47)	< 0.001	1.52 (0.67-3.51)	0.32
HIV status of regular male par	tner					
No regular partner	42.5	44.5	Ref.		Ref.	
HIV-negative	45.1	38.4	0.81 (0.57-1.17)	0.26	1.04 (0.69–1.56)	0.84
Untested/unknown status	7.5	4.3	0.54 (0.24–1.21)	0.14	0.93 (0.39-2.20)	0.87
HIV-positive	4.9	12.8	2.51 (1.43-4.40)	0.001	1.70 (0.87-3.32)	0.12
Anal intercourse with casual m	hale partners (past 6 months)					
No partners/no intercourse	43.4	17.1	Ref.		Ref.	
Consistent condom use	24.5	23.2	2.41 (1.45-4.01)	0.001	1.75 (0.81-3.78)	0.16
Any anal intercourse without condoms	32.1	59.8	4.73 (3.04–7.37)	< 0.001	2.48 (1.18-5.21)	0.02
Ever received post-exposure pr	cophylaxis (PEP)					
No	84.5	70.1	Ref.		Ref.	
Yes	15.5	29.9	2.33 (1.61-3.38)	< 0.001	1.66 (1.09-2.52)	0.02
Personal experience in using c	ondoms*					
Negative/neutral (score < 4)	93.5	97.6	Ref.		Ref.	
Positive (score $\geq 4$ )	6.5	2.4	0.36 (0.13-0.99)	0.05	0.49 (0.17-1.43)	0.19

CI confidence interval, OR odds ratio, p probability, SD standard deviation

\* 9-item scale scored from 1 to 5

health benefits of treatment but not the use of treatment to prevent transmission [10, 19]. Belief in the effectiveness of TasP has increased over the last few years, particularly among HIV-positive men, suggesting attempts to explain the preventative benefits of HIV treatment have had impact. However, there is now a large gap in attitudes between HIV-positive men and their HIV-negative and untested peers, with HIV-positive men holding much more positive views of TasP. This may be a consequence of HIV-positive men being more exposed to and engaged with HIV treatment information and being more open to a strategy that challenges notions of HIV and infectiousness. It remains unclear whether such a gap in knowledge and comfort with TasP is socially divisive, with HIV-positive

	Disagreed/neutral $(n = 309), n (\%)$	Agreed (n = 942), n (%)	Unadjusted OR (95 % CI)	р	Adjusted OR (95 % CI)	р
Mean age in years (SD)	38.7 (12.0)	32.0 (11.5)	0.96 (0.95-0.97)	< 0.001	0.96 (0.95-0.98)	< 0.001
HIV status						
HIV-negative	79.0	79.2	Ref.		Ref.	
Untested/unknown status	7.4	14.0	1.88 (1.18-2.99)	0.008	1.52 (0.92-2.52)	0.11
HIV-positive	13.6	6.7	0.50 (0.33-0.75)	0.001	0.77 (0.49-1.21)	0.27
State or territory						
New South Wales	28.8	35.2	Ref.		Ref.	
Queensland	21.7	18.5	0.70 (0.48-1.00)	0.05	0.58 (0.40-0.86)	0.006
Victoria	27.5	23.6	0.70 (0.50-0.99)	0.04	0.66 (0.46-0.94)	0.02
Other states/territories	22.0	22.7	0.84 (0.59-1.21)	0.35	0.70 (0.48-1.02)	0.06
Employment						
Full-time	58.3	59.3	Ref.		Ref.	
Part-time	11.3	11.5	0.99 (0.66-1.51)	0.98	0.98 (0.63-1.51)	0.91
Student	7.1	16.7	2.30 (1.43-3.70)	0.001	1.41 (0.85–2.33)	0.19
Unemployed/retired/other	23.3	12.5	0.53 (0.38-0.74)	< 0.001	0.64 (0.45-0.92)	0.02
HIV status of regular male pa	artner					
No regular partner	40.5	43.5	Ref.		Ref.	
HIV-negative	42.7	44.7	0.97 (0.74-1.29)	0.84	0.99 (0.74–1.33)	0.95
Untested/unknown status	7.8	6.9	0.83 (0.50-1.37)	0.46	0.60 (0.35-1.03)	0.07
HIV-positive	9.1	4.9	0.50 (0.30-0.83)	0.008	0.70 (0.41-1.22)	0.21

Table 3 Factors associated with agreement that early HIV treatment is necessary (in 2015)

CI confidence interval, M mean, OR odds ratio, p probability, SD standard deviation

men increasingly confident that HIV treatment prevents transmission and most HIV-negative and untested men remaining unconvinced [7, 26]. It is far from clear how a greater reliance on TasP by HIV-positive men will be taken into account by state and territory laws about HIV status disclosure and transmission. These laws tend to specify that 'reasonable precautions' must be taken to prevent HIV transmission, but they do not specify whether having an undetectable viral load is considered 'reasonable' [27]. Related issues have been alluded to in research in the Netherlands, in which it was suggested that HIV-positive men may not disclose their HIV status prior to condomless sex when they have an undetectable viral load [28], troubling the notion of shared responsibility for prevention and potentially fuelling HIV stigma between casual sexual partners [29, 30].

The current gap in belief about TasP may close over time, and it may have little or no impact on sexual practices. However, our results highlight the need to assess both the intended and unintended consequences of antiretroviral-based HIV prevention, using both quantitative and qualitative research methods [31, 32].

## Conclusion

In conclusion, we have found that Australian gay and bisexual men's attitudes to TasP and the early initiation of treatment have become more positive over time. HIVpositive men are much more likely to believe in the effectiveness of TasP than HIV-negative and untested men—a difference that has widened over the last 2 years. We recommend continued monitoring of the social and behavioural effects of the promotion of HIV treatment as prevention.

Acknowledgments We thank the survey participants for their time and Susan Kippax (who advised on the conduct of the study but did not participate in the writing of this article). The Centre for Social Research in Health (CSRH), National Drug Research Institute and Australian Research Centre in Sex, Health and Society receive funding from the Australian Government Department of Health. ACON and Positive Life NSW receive funding from the New South Wales (NSW) Ministry of Health. In 2013 the PrEPARE Project was supported by a research promotion grant from CSRH and in 2015 by the Blood borne viruses and sexually transmissible infections Research, Intervention and Strategic Evaluation (BRISE) program, funded by the NSW Ministry of Health.

#### References

- Cohen MS, Chen YQ, McCauley M, Gamble T, Hosseinipour MC, Kumarasamy N, et al. Prevention of HIV-1 infection with early antiretroviral therapy. N Engl J Med. 2011;365(6):493–505.
- 2. Montaner JSG. Treatment as prevention—a double hat-trick. Lancet. 2011;378(9787):208–9.
- Rodger A, Bruun T, Cambiano V, Vernazza P, Estrada V, Van Lunzen J, et al. HIV transmission risk through condomless sex if HIV+ partner on suppressive ART: PARTNER study. Conference on Retroviruses and Opportunistic Infections; Mar 6; Boston, USA 2014.
- Baggaley R, Doherty M, Ball A, Ford N, Hirnschall G. The strategic use of antiretrovirals to prevent HIV infection: a converging agenda. Clin Infect Dis. 2015;60(suppl 3):S159–60.
- Cairns G, Baker AC, Dedes N, Zakowicz A, West B. Treatment as prevention: arriving at community consensus. Clin Infect Dis. 2014;59(suppl 1):S35–40.
- 6. World Health Organization. Consolidated guidelines on the use of antiretroviral drugs for treating and preventing HIV infection: recommendations for a public health approach. Geneva: World Health Organization; 2013.
- 7. Holt M. HIV pre-exposure prophylaxis and treatment as prevention: a review of awareness and acceptability among men who have sex with men in the Asia-Pacific region and the Americas. Sex Health. 2014;11(2):166–70.
- Young I, McDaid L. How acceptable are antiretrovirals for the prevention of sexually transmitted HIV? A review of research on the acceptability of oral pre-exposure prophylaxis and treatment as prevention. AIDS Behav. 2014;18(2):195–216.
- Rodger A, Phillips A, Speakman A, Gilson R, Fisher M, Wilkins E, et al. Attitudes of people in the UK with HIV who are antiretroviral (ART) naïve to starting ART at high CD4 counts for potential health benefit or to prevent HIV transmission. PLoS ONE. 2014;9(5):e97340.
- Holt M, Lea T, Murphy DA, Ellard J, Rosengarten M, Kippax SC, et al. Australian gay and bisexual men's attitudes to HIV treatment as prevention in repeated, national surveys, 2011–2013. PLoS ONE. 2014;9(11):e112349.
- 11. Carter A, Lachowsky N, Rich A, Forrest JI, Sereda P, Cui Z, et al. Gay and bisexual men's awareness and knowledge of treatment as prevention: findings from the Momentum Health Study in Vancouver, Canada. J Int AIDS Soc. 2015;. doi:10.7448/IAS.18. 1.20039.
- Newman CE, de Wit J, Persson A, Holt M, Slavin S, Kidd MR, et al. Understanding concerns about treatment-as-prevention among people with HIV who are not using antiretroviral therapy. AIDS Behav. 2015;19(5):821–31.
- Persson A. "The world has changed": pharmaceutical citizenship and the reimagining of serodiscordance among couples with mixed HIV status in Australia. Sociol Health Illn. 2015;. doi:10. 1111/1467-9566.12347.
- Commonwealth of Australia. Seventh National HIV Strategy 2014–2017. Canberra: Australian Government Department of Health; 2014.
- 15. NSW Ministry of Health. NSW HIV strategy 2012–2015: a new era. Sydney: NSW Ministry of Health; 2012.
- Health Queensland. Queensland HIV strategy 2013–2015. Brisbane: Queensland Department of Health; 2013.
- Australasian Society for HIV Medicine. Australian Commentary on the US Department of Health and Human Services Guidelines

for the use of Antiretroviral Agents in HIV-1-Infected Adults and Adolescents. Sydney: Australasian Society for HIV Medicine; 2015. http://arv.ashm.org.au/.

- Mao L, Adam PCG, Kippax S, Crooks L, Post JJ, Kidd MR, et al. Evolving views and practices of antiretroviral treatment prescribers in Australia. Med J Aust. 2015;202(5):258–60.
- Holt M, Murphy D, Callander D, Ellard J, Rosengarten M, Kippax S, et al. HIV-negative and HIV-positive gay men's attitudes to medicines, HIV treatments and antiretroviral-based prevention. AIDS Behav. 2013;17(6):2156–61.
- 20. Zablotska I, Kippax S, Grulich A, Holt M, Prestage G. Behavioural surveillance among gay men in Australia: methods, findings and policy implications for the prevention of HIV and other sexually transmissible infections. Sex Health. 2011;8(3):272–9.
- 21. Holt M, Murphy DA, Callander D, Ellard J, Rosengarten M, Kippax SC, et al. Willingness to use HIV pre-exposure prophylaxis and the likelihood of decreased condom use are both associated with unprotected anal intercourse and the perceived likelihood of becoming HIV positive among Australian gay and bisexual men. Sex Transm Infect. 2012;88:258–63.
- Chen Y. Treatment-related optimistic beliefs and risk of HIV transmission: a review of recent findings (2009–2012) in an era of treatment as prevention. Curr HIV/AIDS Rep. 2013;10(1):79–88.
- Poynten IM, Smith DE, Cooper DA, Kaldor JM, Grulich AE. The public health impact of widespread availability of nonoccupational postexposure prophylaxis against HIV. HIV Med. 2007;8(6): 374–81.
- 24. Grulich AE, de Visser RO, Badcock PB, Smith AMA, Heywood W, Richters J, et al. Homosexual experience and recent homosexual encounters: the Second Australian Study of Health and Relationships. Sex Health. 2014;11(5):439–50.
- Poynten IM, Jin FY, Mao LM, Prestage GP, Kippax SC, Kaldor JM, et al. Nonoccupational postexposure prophylaxis, subsequent risk behaviour and HIV incidence in a cohort of Australian homosexual men. AIDS. 2009;23(9):1119–26.
- 26. Holt M, Lea T, Mao L, Zablotska I, Prestage G, de Wit J. HIV prevention by Australian gay and bisexual men with casual partners: the emergence of undetectable viral load as one of a range of risk reduction strategies. J Acquir Immune Defic Syndr. 2015;70(5):545–8.
- Australian Federation of AIDS Organisations. Gay men & HIV disclosure. Sydney: Australian Federation of AIDS Organisations; 2014.
- Van Den Boom W, Stolte I, Witlox R, Sandfort T, Prins M, Davidovich U. Undetectable viral load and the decision to engage in unprotected anal intercourse among HIV-positive MSM. AIDS Behav. 2013;17(6):2136–42.
- Davis M. The 'loss of community' and other problems for sexual citizenship in recent HIV prevention. Social Health Illn. 2008;30(2): 182–96.
- Smit PJ, Brady M, Carter M, Fernandes R, Lamore L, Meulbroek M, et al. HIV-related stigma within communities of gay men: a literature review. AIDS Care. 2012;24(4):405–12.
- Imrie J, Elford J, Kippax S, Hart GJ. Biomedical HIV prevention—and social science. Lancet. 2007;370(9581):10–1.
- 32. de Wit JBF, Adam PCG. Revolution or evolution? What can approaches based on the use of antiretroviral drugs contribute to HIV prevention in gay communities in high-income countries? In: Eaton LA, Kalichman SC, editors. Biomedical advances in HIV prevention. New York: Springer; 2014. p. 181–204.